HONG KONG UNIVERSITY OF SCIENCE & TECHNOLOGY

COMP 102 : Pr ogramming & Computer Fundamentals I

Summer 1996

Laboratory Assignment 3

15 & 17 July

Problem description:

Given three numbers (a, b & c), the middle number (b) is considered to be a peak if it is greater than the number before it (a) and the number after it (c). Similarly, b is considered to be a pit if it is smaller than a and c. Write 1. a function peak which takes three integer arguments (a, b, c) and returns true (1) if b is a peak and false (0) otherwise, and 2. a function pit which takes three integer arguments (a, b, c) and returns true (1) if b is a pit and false (0) otherwise.

In testing the functions peak and pit, 1. write a program which reads in three integers and prints out the second number if it is a peak or pit.

Example : Enter three number: 10 8 9 8 is a pit.

2. write a program to analyze an input Hang Seng Index sequence. Each number in the sequence is an integer representing the closing index of the day. The sequence is terminated by a negative integer. Write a program to find peaks and pits in the input sequence. (The last negative numbers should not be used to define neither a peak nor a pit.)

Example : Input sequence: 9377 9302 9340 9402 9450 9337 9330 9385 9683 9636 -1 9302 is a pit 9450 is a peak 9330 is a pit 9683 is peak

You are required to hand in 1. a design of the algorithm for solving the problem.

2. an implement of your solution using C++. Use separate files for the functions and the main program.